Cardiovascular safety trials of incretin-based drugs: What do they mean?

Incretin-based dipeptidyl peptidase-4 inhibitors and glucagon-like peptide-1 receptor agonists are newer choices of antidiabetic medications that are now most widely used worldwide. Preclinical study results suggest that the two drugs potentially exert benefits to prevent onsets and/or progressions of diabetes-related complications, such as myocardial infarctions and strokes. Outcomes of five clinical trials to evaluate the cardiovascular (CV) safety of dipeptidyl peptidase-4 inhibitors and glucagonlike peptide-1 receptor agonist have been recently reported. The heart failure findings of the Saxagliptin Assessment of Vascular Outcomes Recorded in Patients with Diabetes Mellitus-Thrombolysis in Myocardial Infarction 53 (SAVOR-TIMI53) are unexpected and very concerning; results of the Examination of Cardiovascular Outcomes with Alogliptin vs Standard of Care (EXAMINE), the Trial Evaluating Cardiovascular Outcomes with Sitagliptin (TECOS) and the Evaluation of Lixisenatide in Acute Coronary Syndrome (ELIXA) encourage neutral CV safety profiles of incretin-based drugs in individuals with type 2 diabetes and established CV diseases or multiple CV risks. Furthermore, the Liraglutide Effect and Action in Diabetes: Evaluation of Cardiovascular Outcome Results (LEADER) results show the benefits of liraglutide in preventing

CV events in a similar study population. Despite the many preclinical studies showing the beneficial effects of incretin-related drugs, most CV safety trials of incretin-based drugs, except for LEADER, did not show benefits for CV events. It is important to recognize that CV safety trials were carried out to meet the US Food and Drug Administration guidance to assess CV safety of all new antidiabetic drugs; they were not designed to assess their benefits for CV events. Therefore, the long-term potential benefit, as well as even the safety, of incretin-based drugs for certain CV outcomes has not been definitively established, and requires evaluation in more specific and more relevant trials. If the need for CV safety trials would be determined based on an individual drug's safety data during its earlier development as well as its mechanism of action, resources could be saved for carrying out such clinical trials.

Chronic hyperglycemia, in collaboration with hypertension and dyslipidemia, can cause diabetes-associated microvascular complications (e.g., neuropathy, nephropathy and retinopathy) and macrovascular complications (e.g., myocardial infarctions, strokes and peripheral arterial diseases) in individuals with diabetes. Lines of evidence show that amelioration of glycemia with appropriate controls of bodyweight, blood pressures, and lipid levels prevents onset and/or progression of such complications. To date, several glucose-lowering drugs have been developed to normalize glycemia in individuals with type 2 diabetes. Among such drugs, incretin-based dipeptidyl

peptidase-4 inhibitors (DPP-4is) and glucagon-like peptide-1 receptor agonists (GLP-1RAs) are newer choices of such antidiabetic medications. The two drugs are now most widely used worldwide, in part because they have low risks of hypoglycemia and bodyweight gain despite their ability to ameliorate glycemia through enhancement of insulin secretion, unlike sulfonylureas and glinides¹. DPP-4is improve glycemic control in individuals with type 2 diabetes by preventing degradation of the two incretins, glucagon-like peptide-1 (GLP-1) and glucose-dependent insulinotropic polypeptide. GLP-1RAs does so by binding to the GLP-1 receptor and activating GLP-1 receptor signaling. GLP-1 and glucosedependent insulinotropic polypeptide are secreted from the intestine on ingestion of various nutrients and enhance insulin secretion from pancreatic β-cells glucosedependently. Preclinical studies in animal models have shown diverse biological functions of both incretins in addition to their glucose-dependent insulinotropic action². Thus, it has been expected that the incretin-related drugs potentially exert benefits to prevent onsets and/or progressions of diabetes-related complications, such as myocardial infarctions (MI) and strokes. However, the effects of incretin-based drugs on diabetes-related complications need to be examined in clinical trials with adequately powered, prospective, controlled relevant endpoints. For these reasons, outcomes of five clinical trials to evaluate the cardiovascular (CV) safety of individual incretin-based drugs have gained much attention.

Three trials, the Saxagliptin Assessment of Vascular Outcomes Recorded in Patients with Diabetes Mellitus-Thrombolysis in Myocardial Infarction 53 (SAVOR-TIMI53), the Examination of Cardiovascular Outcomes with Alogliptin vs Standard of Care (EXAMINE) and

^{*}Corresponding author. Daisuke Yabe Tel: +81-75-751-3560 Fax: +81-75-751-4244 E-mail address: ydaisuke-kyoto@umin.ac.jp Received 2 September 2016; revised 6 September 2016; accepted 7 September 2016

^{© 2016} The Authors. Journal of Diabetes Investigation published by Asian Association for the Study of Diabetes (AASD) and John Wiley & Sons Australia, Ltd This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

the Trial Evaluating Cardiovascular Outcomes with Sitagliptin (TECOS), assessed CV safety of the DPP-4is saxagliptin, alogliptin and sitagliptin in individuals with type 2 diabetes at risk for CV events, respectively. SAVOR-TIMI53 was carried out globally using a total of 16,492 patients with a history of CV disease (approximately 80% of the study population) or with multiple CV risks (approximately 20%) (Table 1)³. The median observation period was 2.1 years; glycated hemoglobin (HbA1c) changes from baseline were just 0.3% greater in those receiving saxagliptin compared with a placebo. The primary composite end-point of CV death, non-fatal MI and non-fatal ischemic stroke occurred in patients receiving saxagliptin similarly to those receiving a placebo (hazard ratio [HR] 1.00, 95% confidence interval [CI] 0.89-1.12, P = 0.99). EXAMINE was carried out globally using a total of 5,380 patients, all of whom had acute coronary syndrome⁴. The median observation period was 1.6 years; HbA1c changes from baseline were 0.4% greater in those receiving alogliptin. The primary composite end-point of CV death, non-fatal MI and non-fatal stroke occurred in patients receiving alogliptin similarly to those receiving a placebo (HR 0.96, upper 99% CI: <1.16, P = 0.32). TECOS was carried out globally using a total of 14,671 patients with a history of CV disease (approximately 75%), ischemic stroke (approximately 25%) and/or peripheral artery diseases (approximately 20%)⁵. The median observation period was 3.0 years; HbA1c changes from baseline were 0.3% greater in those receiving sitagliptin. The primary composite endpoint of CV death, non-fatal MI, nonfatal stroke and hospitalization for unstable angina was 0.98 (95% CI: 0.89-1.08, P = 0.65). These results can be taken to show that DPP-4is have neutral CV safety profiles in individuals with type 2 diabetes and high risks for CV events, particularly MI, stroke and CV death. Unexpectedly, saxagliptin use was associated with a significant increase in hospitalization for heart failure compared with a placebo (HR 1.27, 95% CI: 1.07-1.51, P = 0.007). EXAMINE and TECOS did not find significant associations of heart failure with use of alogliptin and sitagliptin, suggesting that the effects of saxagliptin on heart failure might be a drug effect rather than a class effect. Rigorous post-hoc analyses on SAVOR-TIMI53 failed to identify obvious causal mechanisms for heart failure with saxagliptin; further investigations are warranted to evaluate changes in cardiac function in individuals with type 2 diabetes receiving saxagliptin.

Two clinical trials, the Evaluation of Lixisenatide in Acute Coronary Syndrome (ELIXA) and the Liraglutide Effect and Action in Diabetes: Evaluation of Cardiovascular Outcome Results (LEADER), assessed the CV safety of the GLP-1RAs lixisenatide and liraglutide in individuals with type 2 diabetes at risk for CV events, respectively. ELIXA was carried out globally using a total of 6,068 patients with a recent experience of acute coronary syndrome (Table 1)⁶. The median observation period was 2.1 years; HbA1c changes from baseline were approximately 0.3% greater in those receiving lixisenatide compared with a placebo. The primary composite endpoint of CV death, non-fatal MI, nonfatal stroke and hospitalization for unstable angina occurred in patients receiving lixisenatide similarly to those receiving placebo (HR 1.02, 95% CI: 0.89-1.17, P = 0.81). LEDER was carried out globally using a total of 9,340 patients with high risks for CV events⁷. The median observation period was 3.8 years; HbA1c changes from baseline were approximately 0.4% greater in those receiving liraglutide compared with a placebo. The primary composite end-point of CV death, non-fatal MI and non-fatal stroke occurred in significantly fewer patients receiving liraglutide (HR 0.87, 95% CI: 0.78–0.97, P < 0.001 for non-inferiority and P = 0.01 for superiority). Although the differing outcomes of ELIXA and LEADER are still unexplained, differences in baseline characteristics of the enrolled patients, such as HbA1c (LEADER \approx 8.7%/ELIXA \approx 7.7%) and body mass index (LEADER ≈ 32 /ELIXA ≈ 30),

might have implications in accord with differences in the mode of action (i.e., short-acting vs long-acting)⁸. Further investigations are required to understand the factors responsible for the differing outcomes of ELIXA and LEADER. Nevertheless, it is important to consider that hospitalization for heart failure was not affected by lixisenatide use (HR 0.96, 95% CI: 0.75–1.23, P = 0.75) or by liraglutide use (HR 0.87, 95% CI: 0.73–1.05, P = 0.14).

The heart failure findings of SAVOR-TIMI53 are unexpected and very concerning; the results of EXAMINE, TECOS and ELIXA encourage neutral CV safety profiles of incretin-based drugs in individuals with type 2 diabetes and established CV diseases or multiple CV risks. Furthermore, LEADER results show benefits of liraglutide for preventing CV events in a similar study population. Because of the existence of the many preclinical studies showing the beneficial effects of incretins and incretin-related drugs, obvious questions arise as to why most CV safety trials of incretin-based drugs, except for liraglutide, did not show benefits for CV events. In this regard, it is important to recognize that CV safety trials were carried out to meet the US Food and Drug Administration guidance implemented in 2008 to assess the CV safety of all new antidiabetic drugs; they were not designed to assess their benefits for CV events. CV safety trials require a large study population in multiple countries to obtain CV events sufficient for statistical analysis. Monitoring such large study populations globally requires substantial costs and efforts; thus encouragenrollment of patients with ing established CV diseases who are likely to experience a CV event within a limited time-period. These patients are already taking aspirin, statin and angiotensinconverting enzyme inhibitors/angiotensin II receptor blockers along with multiple antidiabetic drugs (Table 1); the results obtained should be evaluated taking these patient characteristics into consideration. Therefore, the long-term potential benefit, as well as even the safety, of incretinbased drugs for certain CV outcomes has

2

	UKPDS33	SAVOR-TIMI53	EXAMINE	TECOS	ELIXA	LEADER
n Follow …n (værs)⁺	1 2,729/C 1,138 100	D 8,280/P 8,212 2 1	D 2,701/P 2,679 1.6	D 7,332/P 7,339 3.0	D 3,034/P3,034 2 1	D 4,668/P 4,672 3
Primary end-point	Any diabetes-related end-points	CV death, non-fatal MI and non-fatal ischemic stroke	CV death, non-fatal MI and non-fatal stroke	CV death, non-fatal MI, non-fatal stroke and	CV death, non-fatal MI, nonfatal stroke and hospitalization for	CV death, non-fatal MI and non-fatal stroke
				hospitalization for	unstable	
	HR 0.88 (0.79–0.99)	HR 1.00 (0.89–1.12)	HR 0.96 (upper	UI 31400 AN	angina HR 1.02 (0.89–1.17)	HR 0.87 (0.78-0.97)
	P = 0.029	P = 0.99 P < 0.001	boundary of	P < 0.001 for	P < 0.001	P < 0.001 for
		for non-inferiority	the one-sided	non-inferiority HR	for non-inferiority	non-inferiority $P = 0.01$
		F = 0.99 for superiority	P < 0.001 for	P = 0.65 for	F = U.S.I for superiority	ior superiority
			non-inferiority D = 0.32 for	superiority	-	
			r — 0.32 IUI superiority			
Death from	ND	HR 1.03 (0.87–1.22) P = 0.72	HR 0.79 (0.60–1.04)	HR 1.03 (0.89–1.19)	HR 0.98 (0.78–1.22)	HR 0.78 (0.66–0.93)
CV cause			P = 0.10	P = 0.71	P = 0.85	P = 0.007
M	HR 0.84 (0.71–1.00)	HR 0.95 (0.80–1.12) $P = 0.52$	HR 1.08 (0.88–1.33)	HR 0.95 (0.81–1.11)	HR 1.03 (0.87–1.22)	HR 0.86 (0.73–1.00)
	P = 0.052		P = 0.47	P = 0.49 (fatal	P = 0.71	P = 0.046 (fatal or
-			(non-fatal)	or non-fatal)	(fatal or non-fatal)	non-fatal)
Stroke	HR 1.11 (0.81–1.51)	HR 1.11 ($0.88-1.39$) $P = 0.38$	HR 0.91 (0.55–1.50)	HR 0.97 (0.79–1.19)	HR 1.12 (0.79–1.58)	HR 0.86 (0.71–1.06)
	P = 0.52	(ischemic)	P = 0.71	P = 0.66 (fatal	P = 0.54	P = 0.16 (tatal or
			(non-fatal)	or non-fatal)	(fatal or non-fatal)	non-fatal)
Hospitalization		HK = 1.21 (1.0.1 - 1.0.1) / 2.1	HK 1.0/ (0./9–1.46)	HK 1.00 (0.83–1.20)	HK 0.96 (0./5-1.23) 0 0.75	HK 0.8/ $(0./3-1.05) = 9.14$
TOT HF			$\gamma = 0.00$	F = 0.98	V = 0.70	
Age (years)*	23.2 ± 8.0/C 23.4 ± 8.0	0.5 ± 0.co 7/c.8 ± 1.co U	U 61.0/P 61.0	D 65:4 ± 7.9/ P 65.5 + 80	Р КЛК + 9К Р КЛК + 9К	D 04.2 エ / .2/ド 04.4 エ / .2
Sex male (%)	1606/0619	D 666/P 673	D 67 7/P 680	0.2 T CLU 1	D 696/P 691	D 645/P 640
Ethnicity, white (%)	181/C 81	D 75.4/P75.1	D 72.8/P72.5	D 67.6/P 68.2	D 74.4/P 76.4	D 77.5/P 77.5
CV disease (%)	Ι	D 78.4/P 78.7	D 100/P100	D 73.6/P 74.5	D 100/P 100	D 82.1/P 80.6
Duration of	Newly diagnosed	D 10.3 (5.2–16.7)/P	D 7.1 (2.6–13.8)/	D 11.6 ± 8.1/	D 9.2 ± 8.2/	D 12.8 ± 8.0/P 12.9 ± 8.1
diabetes (years) [§]		10.3 (5.3–16.6)	P 7.3 (2.8–13.7)	P 11.6 ± 8.1	P 9.4 ± 8.3	
HbA1c (%)	7.09 ± 1.54/C 7.05 ± 1.42	D 8.0 ± 1.4/P 8.0 ± 1.4	D 8.0 ± 1.1/P 8.0 ± 1.1	D 7.2 ± 0.5/	D 7.7 ± 1.3/	D 8.7 ± 1.6/P 8.7 ± 1.5
i				P 7.2 ± 0.5	P 7.6 ± 1.3	
BMI	27.5 ± 5.1/C 27.8 ± 5.5	D 31.1 ± 5.5/P 31.2 ± 5.7	D 28.7 (15.7-55.9)/ D 28.7 (15.6 68.3)	D 30.2 ± 5.6/ P 30.2 + 5.7	D 30.1 ± 5.6/ P 30.2 + 5.8	D 32.5 ± 6.3/P 32.5± 6.3
Matterion (02)						D 75 8/D 770
sulfonvlurea (%)		D 40.5/P 40.0	D 46.9/P 46.2	D 45.6/P 45.0	D 32.6/P 33.5	D 50.6/P 50.5
Insulin (%)	1 0/C 0	D 41.6/P 41.2	D 29.4/P 30.3	D 23.5/P 22.9	D 39.2/P 39.0	D 43.6/P 45.5
Aspirin (%) ^{††}	11.7/C1.5	D 75.5/P 75.0	D 90.6/P 90.8	D 78.6/P 78.4	D 97.6/P97.4	D 68.6/P66.8
C++: /0/\11						

	UKPDS33	SAVOR-TIMI53	EXAMINE	TECOS	ELIXA	LEADER
β- Blocker (%) ARB (%) ACEI (%)	I 12/C 12	D 61.6/P 61.6 D 28.2/P 27.6 D 53.6/P 54.9	D 81.7/P 82.2 D 81.5/P 82.5	D 63.4/P 63.7 D 20.1/P 21.1 D 58.2/P 58.1	D 83.6/P85.3 D 84.9/P 85.0	D 56.7/P54.0 D 31.8/P 31.8 D 51.7/P 50.3
[†] Median. [‡] Mean ± standard except for median with ran- and EXAMINE. [¶] Mean ± star of aspirin in the Evaluation . ^{‡‡} Use of Ilpid-lowering drug blockers; BMI, body mass in not determined; P, placebo.	E standard deviation except n with range in the Saxaglip ean ± standard deviation ex ivaluation of Lixisenatide in v ering drugs, instead of statir ty mass index; C, conventior 7, placebo.	[†] Median. [‡] Mean ± standard deviation except for median in the Examination of Cardiovascular Outcomes with Alogliptin vs Standard of Care (EXAMINE). [§] Mean ± standard deviation except for median with range in the Saxagliptin Assessment of Vascular Outcomes Recorded in Patients with Diabetes Mellitus-Thrombolysis in Myocardial Infarction 53 (SAVOR-TIMIs and EXAMINE. [¶] Mean ± standard deviation except for median with range in the Trial Evaluating Cardiovascular Outcomes with Sitagliptin (TECOS). ^{††} Use of antiplatelet agent (%) inst of aspirin in the Evaluation of Lixisenatide in Acute Coronary Syndrome (ELIXA) and Liraglutide Effect and Action in Diabetes: Evaluation of Cardiovascular Outcome Results (LEADER). ^{‡†} Use of lipid-lowering drugs, instead of statin in the United Kingdom Prospective Diabetes Study (UKPDS). ACEI, angiotensin-converting enzyme inhibitors; ARB, angiotensin II-recept blockers; BMI, body mass index; C, conventional therapy, CV, cardiovascular; D, drug of interest, HbA1C, glycated hemoglobin; HF, heart failure; HR, hazard ratio; I, intensive therapy; NI not determined; P, placebo.	of Cardiovascular Outcom comes Recorded in Patien the Trial Evaluating Cardic (A) and LingJutide Effect a ective Diabetes Study (UK) O, drug of interest, HbA1c,	les with Alogliptin vs Sta ts with Diabetes Mellitus vvascular Outcomes with and Action in Diabetes: E PDS). ACEI, angiotensin-cc glycated hemoglobin; H	ndard of Care (EXAMINE -Thrombolysis in Myocar Sitagliptin (TECOS). ⁴¹ Us valuation of Cardiovascu onverting enzyme inhibii F, heart failure; HR, haza	[†] Median. [‡] Mean ± standard deviation except for median in the Examination of Cardiovascular Outcomes with Alogliptin vs Standard of Care (EXAMINE). [§] Mean ± standard deviation except for median with range in the Saxagliptin Assessment of Vascular Outcomes Recorded in Patients with Diabetes Mellitus-Thrombolysis in Myocardial Infarction 53 (SAVOR-TIMIS3) and EXAMINE. [¶] Mean ± standard deviation except for median with range in the Saxagliptin Assessment of Vascular Outcomes Recorded in Patients with Diabetes Mellitus-Thrombolysis in Myocardial Infarction 53 (SAVOR-TIMIS3) and EXAMINE. [¶] Mean ± standard deviation except for median with range in the Trial Evaluating Cardiovascular Outcomes with Sitagliptin (TECOS). ^{††} Use of antiplatelet agent (%) instead of aspirin in the Evaluation of Lixisenatide in Acute Coronary Syndrome (ELIXA) and Liraglutide Effect and Action in Diabetes: Evaluation of Cardiovascular Outcome Results (LEADER). [‡] Use of lipid-lowering drugs, instead of statin in the United Kingdom Prospective Diabetes Study (UKPDS). ACE, angiotensin-converting enzyme inhibitors; ARB, angiotensin II-receptor blockers; BMI, body mass index; C, conventional therapy, CV, cardiovascular; D, drug of interest, HbA1c, glycated hemoglobin; HF, heart failure; HR, hazard ratio; I, intensive therapy; ND, not determined; P, placebo.

not been definitively established and requires evaluation of individuals without CV diseases, similar to what was done in the United Kingdom Prospective Diabetes Study (UKPDS)9, making full use of appropriate surrogate markers (e.g., intima-media thickness on the carotid ultrasonography). In this respect, the question remains whether the US Food and Drug Administration guidance for assessing CV safety through large CV safety trials of every new antidiabetes drug is appropriate. If the need for CV safety trials would be determined based on an individual drug's safety data during its earlier development as well as its mechanism of action, resources could be saved for carrying out more specific and more relevant trials (e.g., prevention of onset and/or progression of diabetesrelated micro- and macrovascular complications) or trials to find effective therapeutics for reducing CV death and improving long-term healthy survival.

ACKNOWLEDGMENTS

D Yabe and Y Seino contributed to intellectual discussions and writing the manuscript. D Yabe and Y Seino are the guarantors of this work.

DISCLOSURE

D Yabe received consulting and/or speaker fees from MSD K.K. and Novo Nordisk Pharma Ltd. D Yabe received commissioned/joint clinical research grants from Nippon Boehringer Ingelheim Co. Ltd., Eli Lilly and Company, Taisho Toyama Pharmaceutical Co. Ltd. and MSD K.K, Takeda Pharmaceutical Company Limited, Ono Pharmaceutical Co. Ltd., Novo Nordisk Pharma Inc. and Arklay Co. Ltd. Y Seino received consulting and/or speaker fees from Eli Lilly Japan K.K., Sanofi K.K., Novo Nordisk Pharma Inc., Glaxo-Smith-Kline, Taisho Pharmaceutical Co. Ltd., Taisho Toyama Pharmaceutical Co. Ltd., Astellas Pharma Inc., BD, Nippon Boehringer Ingelheim Co. Ltd., Johnson & Johnson and Takeda Pharmaceutical Company Limited. Y Seino received clinical commissioned/ joint research grants from Taisho Toyama Pharmaceutical Co. Ltd., Nippon

Boehringer Ingelheim Co. Ltd., Eli Lilly and MSD K.K.

Daisuke Yabe^{1,2,3}*, Yutaka Seino^{1,4} ¹Yutaka Seino Distinguished Center for Diabetes Research, Kansai Electric Power Medical Research Institute, Kobe, ²Department of Diabetes, Endocrinology and Nutrition, Kyoto University Graduate School of Medicine, Kyoto, ³Division of Molecular and Metabolic Medicine, Kobe University Graduate School of Medicine, Kobe, and ⁴Center for Diabetes, Endocrinology and Metabolism, Kansai Electric Power Hospital, Osaka, Japan

REFERENCES

- 1. Seino Y, Kuwata H, Yabe D. Incretinbased drugs for type 2 diabetes: focus on East Asian perspectives. *J Diabetes Investig* 2016; 7(Suppl 1): 102– 109.
- 2. Seino Y, Yabe D. Glucose-dependent insulinotropic polypeptide and glucagon-like peptide-1: incretin actions beyond the pancreas. *J Diabet Investig* 2013; 4: 108–130.
- 3. Scirica BM, Bhatt DL, Braunwald E, et al. Saxagliptin and cardiovascular outcomes in patients with type 2 diabetes mellitus. *N Engl J Med* 2013; 369: 1317–1326.
- 4. White WB, Cannon CP, Heller SR, *et al.* Alogliptin after acute coronary syndrome in patients with type 2 diabetes. *N Engl J Med* 2013; 369: 1327–1335.
- 5. Green JB, Bethel MA, Armstrong PW, et al. Effect of sitagliptin on cardiovascular outcomes in type 2 diabetes. N Engl J Med 2015; 373: 232– 242.
- 6. Pfeffer MA, Claggett B, Diaz R, *et al.* Lixisenatide in patients with type 2 diabetes and acute coronary syndrome. *N Engl J Med* 2015; 373: 2247–2257.
- Marso SP, Daniels GH, Brown-Frandsen K, et al. Liraglutide and cardiovascular outcomes in type 2 diabetes. N Engl J Med 2016; 375: 311–322.
- Yabe D, Seino Y. Defining the role of GLP-1 receptor agonists for individualized treatment of type 2

Table 1 (Continued)

diabetes. *Expert Rev Endocrinol Metab* 2014; 9: 659–670.

9. UK Prospective Diabetes Study (UKPDS) Group. Intensive bloodglucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes (UKPDS 33). UK Prospective Diabetes Study (UKPDS) Group. *Lancet* 1998; 352: 837–853.

Doi: 10.1111/jdi.12576